Tick Size Constraints, Market Structure, and Liquidity¹

First Draft: November 18, 2012

This draft: December 28, 2013

Chen Yao and Mao Ye

1,

¹Chen Yao is from the University of Warwick and Mao Ye is from the University of Illinois at Urbana-Champaign. Please send all correspondence to Mao Ye: University of Illinois at Urbana-Champaign, 340 Wohlers Hall, 1206 South 6th Street, Champaign, IL, 61820. E-mail: maoye@illinois.edu. Telephone: 217-244-0474. This paper is the first part of a paper circulated under the title "The Externalities of High Frequency Trading." We thank Jim Angel, Shmuel Baruch, Robert Battalio, Dan Bernhardt, Jonathan Brogaard, Jeffery Brown, John Campbell, John Cochrane, Robert Frank, Slava Fos, George Gao, Paul Gao, Arie Gozluklu, Joel Hasbrouck, Frank Hathaway, Pankaj Jain, Tim Johnson, Charles Jones, Andrew Karolyi, Nolan Miller, Katya Malinova, Maureen O'Hara, Neil Pearson, Richard Payne, Andreas Park, Josh Pollet, Gideon Saar, Ronnie Sadka, Jeff Smith, Duane Seppi, an anonymous reviewer for the FMA Napa Conference, and seminar participants at the University of Illinois, the SEC, CFTC/American University, the University of Memphis, the University of Toronto, NBER market microstructure meeting, the 8th Annual MARC meeting, the Financial Intermediation Research Society Conference 2013, the 3rd MSUFCU Conference on Financial Institutions and Investments, the Northern Finance Association Annual Meeting 2013, the China International Conference in Finance 2013, and the 9th Central Bank Workshop on the Microstructure of Financial Markets for their helpful suggestions. We also thank NASDAQ OMX for providing the research data. This research is supported by National Science Foundation grant 1352936. This work also uses the Extreme Science and Engineering Discovery Environment (XSEDE), which is supported by National Science Foundation grant number OCI-1053575. We thank Robert Sinkovits and Choi Dongju of the San Diego Supercomputer Center and David O'Neal of the Pittsburgh Supercomputer Center for their assistance with supercomputing, which was made possible through the XSEDE Extended Collaborative Support Service (ECSS) program. We also thank Jiading Gai, Chenzhe Tian, Tao Feng, Yingjie Yu, and Chao Zi for their excellent research assistance.

Abstract

We argue that a one-penny minimum tick size for all stocks priced above \$1 (SEC rule

612) encourages high-frequency trading and taker/maker-fee markets. We find that non-high

frequency traders (non-HFTers) are 2.7 times more likely than HFTers to provide best prices,

thereby establishing price priority. However, while relatively large tick sizes constrain non-

HFTers from providing better prices, HFTers' speed advantage helps them establish time priority

over non-HFTers. Because liquidity providers can undercut prices by paying a maker fee, non-

HFTers enter the taker/maker market more frequently than HFTers; this tendency is even

stronger when tick size constraints are high. When stock splits increase relative tick size, speed

competition in liquidity provision increases and volume shifts to the taker/maker market,

although without improving liquidity. Thus recent proposals to increase tick size will not

improve liquidity. Instead, they will generate further speed competition and taker/marker

markets.

Key Words: tick size, high-frequency trading, maker/taker fees, liquidity

2

1. Introduction

Under standard Walrasian equilibrium, price is infinitely divisible but time is not; all agents are assumed to arrive at the market at the same time. However, the reality regarding financial markets is exactly the opposite: price competition is restricted by tick size regulations but time becomes divisible at the nanosecond level. This paper shows that two sources of friction, discrete prices and (almost) continuous time under canonical Walrasian equilibrium, help to explain two important features of the U.S. stock market: high-frequency trading (HFT) and taker/maker fees. Currently, there are heated debates over 1) whether we should increase tick size and 2) whether HFT and taker/maker fees should be regulated and what form such new regulation should take. To the best of our knowledge, our paper is the first to show that HFT and taker/maker fees are the consequences of existing tick size regulations, and we argue that increasing tick size through new regulatory action would further encourage speed competition by discouraging price competition and leading to a proliferation of markets with taker/maker fees.

Tick size in the United States is regulated through SEC rule 612 (Minimum Pricing Increment) of regulation NMS. The rule prohibits stock exchanges from displaying, ranking, or accepting quotations, orders, or indications of interest in any NMS stock priced in an increment smaller than \$0.01 if the quotation, order, or indication of interest is priced equal to or greater than \$1.00 per share.² A recent study by Credit Suisse demonstrates that this one-penny tick size constraint is surprisingly binding: 50 percent of S&P stocks priced below \$100 per share have one-penny quoted spreads (Avramovic, 2012). The clustering of quoted spreads on one penny suggests that many of those stocks should have an equilibrium bid-ask spread of less than one

² There are some limited exemptions such as Retail Price Improvement (RPI) Program and mid-point peg orders.

penny if there are no tick size constraints. The minimum pricing increment rule imposes a price floor on the lowest price for liquidity in the public exchange. A surplus represents a natural response to a binding price floor (supply exceeds demand). Because a price floor prevents the pricing system from rationing the available supply, other mechanisms must take its place. Rockoff (2008) summarizes four possible responses to price controls: queuing, the emerging of markets that bypass the regulation, evading, and rationing. HFT and taker/maker fees are just two examples of this general economic principle. HFT is a form of queuing through which traders with high speed capacity can move to the front of the queue at a constrained price. Taker/maker fees make it possible to bypass tick-size regulations, allowing traders to quote at sub-penny levels.

We first demonstrate that tick-size constraints are among the driving forces of HFT liquidity provision. Displayed limit orders in the NASDAQ market observe price and time priority.³ Among orders with the same price and display status, orders arriving first have the highest priority. Therefore, a large tick size increases the importance of speed competition but discourages price competition. Because SEC rule 612 mandates a one-cent tick size for all stocks, the regulation imposes high tick size constraints for two types of stocks: 1) low-priced stocks, for which a fixed nominal tick size leads to a higher relative tick size, and 2) large stocks, for which the equilibrium bid-ask spread can be below one tick. We find that high-frequency traders (HFTers) with their speed advantage take a higher market share in terms of volume relative to lower-frequency traders in low-priced stocks with large caps.

Tick size constraints and the comparative advantages of non-HFTers and HFTers are the main economic mechanisms driving this cross-sectional variation. We first find that non-HFTers

³ Displayed orders take priority over non-displayed orders when they have the same price. For want of space, we do not discuss order display strategy here. Yao (2013) offers an empirical examination of display vs. hidden orders.

are 2.7 times more likely than HFTers to quote better prices and the likelihood that non-HFTers to provide better prices increases as relative tick size decreases. Therefore, a small tick size helps non-HFTers to establish price priority over HFTers. For large stocks with low prices, or stocks with large relative tick size, non-HFTers are still more likely to quote better prices than HFTers (2.8% vs. 1.9%), but the relatively larger tick size of those stocks apparently imposes a constraint on non-HFTers' ability to undercut HFTers. For this stock category, 95.4% of the time HFTers and non-HFTers both provide the best price, which implies that time is needed as a secondary priority rule for allocating resources. Next, we find that the volume of HFTer liquidity providers is the highest for this category (49.29%), but the number decreases to 35.53% for large stocks with small relative tick sizes. In summary, we find that HFTers are less likely to provide better prices than non-HFTers, but large tick sizes enable them to quote the same price as non-HFTers and their speed helps them establish time priority at constrained prices. Therefore, tick size constraints favor HFTer liquidity provision by encouraging speed competition and discouraging price competition.

Taker/maker fees are another market design response to tick-size constraints. It provides liquidity providers with a means of undercutting prices by paying the stock exchange the maker fee, which is used by the stock exchange to partially subsidize liquidity takers. The impact of tick size constraints is demonstrated using the following identification strategy. Direct Edge, a stock exchange that executes 12% of U.S. equity trading volume, operates twin trading platforms: EDGA and EDGX. These two trading platforms are almost identical except for the fee structure. In our sample period, EDGA, like most other exchanges, uses a maker/taker fee structure, which

-

⁴ This might be either because undercutting is too costly or because there is no room to undercut the price when the bid-ask spread is exactly one penny.

⁵ Internalization, dark pools, mid-point peg orders, and the Retail Price Improvement Program are also exempted from SEC rule 612.

pays liquidity makers 0.26 cents and charges liquidity takers 0.3 cents. EDGA, however, has an inverted taker/maker fee structure whereby the maker of liquidity, or the passive (limit) order, is charged a fee of 0.025 cents and the taker of liquidity, or the aggressive (market) order, obtains a rebate of 0.015 cents. Two interesting questions immediately emerge. First, why are some liquidity providers paid whereas others need to pay when providing liquidity for the same stock? Second, what forces determine the competition for order flow between these two markets?

We find the evidence that non-HFTers enter the taker/maker market more frequently than HFTers, while non-HFTers are more likely to pay the maker fee when tick size constraints are high. This (imperfect) separating equilibrium is generated through the comparative advantage enjoyed by HFTers. Holding all else equal, the unit profit obtained to make the market is higher in the maker/taker market conditional on execution, but such execution needs to be at the front of the queue. Non-HFTers do not have the speed advantage to move to the front of the queue. Retail traders, for example, usually have very low execution probability in maker/taker markets (Battalio, Corwin and Jennings, 2013). However, non-HFTers can choose to pay a fee and enter taker/maker market. For low priced stocks, Non-HFTers rely more on taker/maker market to undercut the price because of tick size constraints. Therefore, non-HFTers are more likely to pay to provide liquidity, whereas HFTers are more likely to get paid by providing liquidity because of their speed advantage. We also find that tick size constraints play a prominent role in determining competition for order flow. Taker/maker market has a surprisingly high market share for stocks with relatively high tick sizes. For low-priced large-cap stocks, EGDA executes 64.28% of the volume leaving the EDGX market share at only 35.72%. As stock prices increase, the impact of EDGA decreases. The result is consistent with the seminal theoretical work of Foucault, Kadan, and Kandel (2013). Their model posits an optimal bid-ask spread, which is the

bid-ask spread that maximizes trading volume. Mandated tick sizes impose a constraint that prevents the bid-ask spread from freely adjusting, but stock exchanges can change taker/maker fees to move the spread to the optimal size. When tick size is too large, charging makers and subsidizing takers (taker/maker model) can increase trading volume. An increase in stock prices decreases relative tick size, which reduces the need to adjust the relative tick size closer to the optimal size by charging makers. Therefore, volume increases in markets with maker/taker fees.

U.S. regulators recently proposed increasing tick sizes, arguing that large ticks improve liquidity, facilitate stock research, and ultimately increase IPOs. 6 MiFID II of Europe also proposes increasing tick sizes with the goal of controlling HFT. We test these two hypotheses using stock splits as exogenous shocks to tick-size constraints. We find that an increase in the relative tick size due to stock splits does not improve liquidity. First, the proportional spread increases because the decrease in the spread is less than the decrease in the nominal stock price. The depth, or the queue of traders able and willing to provide liquidity, increases when the quoted spread increases. With increased quoted spread and depth, the key to comparing liquidity is through the effective spread, or the actual transaction cost to liquidity demanders. We find that the effective spread increases, particularly when liquidity demanders also need to pay the taker fee. Therefore, an increase in the relative tick size does not improve liquidity. Meanwhile, we observe an increase in the speed of liquidity provision, which is consistent with our results using a cross-sectional variation of the relative tick size. Finally, there is a migration of volume from EDGX to EDGA. In summary, an increase in the relative tick size does not improve liquidity, but instead it drives speed competition and the migration of volume to the trading platform that bypasses the tick size constraints.

⁶

⁶ "Exchanges Said to Prepare Pilot Programs for Changing Tick Sizes," *Bloomberg News*, June 13, 2013.

⁷ "MiFID II aims to slow HFT with increase in minimum tick sizes," *Markets Media*, October 10, 2012

Several recent studies also address the tick size issue, including Bartlett and McCray (2013), O'Hara, Saar, and Zhong (2013), and Buti, Consonni, Rindi, and Werner (2013). Our paper is, however, the first to link tick size with both HFT and taker/maker fees, and also the first to empirically examine the economic mechanism that drives the results: HFTers' comparative speed advantage and non-HFTers' comparative price advantage. O'Hara, Saar, and Zhong (2013) and Buti, Consonni, Rindi, and Werner (2013) also examine order flow competition between exchanges and trading venues that can quote sub-penny prices. However, our paper has the advantage of cleaner identification: EDGA and EDGX are identical except for the fee structure, whereas the trading platform on which these other studies are based differs along other dimensions such as information revelation and the trading mechanism.

To the best of our knowledge, our paper is the first empirical study that provides a coherent explanation linking the three streams of literature on tick size, HFT, and taker/maker fees. Tick size constraints create rents for supplying liquidity and an oversupply of liquidity at constrained prices. Traders who achieve higher speeds are able to supply liquidity because of time priority. Tick-size constraints also induce some traders to pay a fee in order to make the market. This explains the proliferation of markets with inverted fees. As expected, HFTer liquidity provision is strongly correlated with the taker/maker market share relative to the maker/taker market share because they are both driven by tick size constraints. By sorting the relevant stocks into five portfolios, we find that the highest EDGA-to-EDGX market-share quintile experiences twice as much HFT market-making activity as the lowest EDGA-to-EGDX market-share quintile.

Our study represents two contributions to the policy debate on tick size, maker/taker fees, and HFT. First, this is the first empirical study that demonstrates the linkage between these three

_

policy issues. 8 Second, instead of discussing whether and how we should regulate HFT and maker/taker fees, our paper is the first to propose that these two phenomena may be a consequence of existing tick size regulations. Economic reasoning and our empirical evidence can show, step by step, how various regulations create these two phenomena. At infinitely small fees would be neutralized by differences in the nominal bid-ask spread: the tick sizes. maker/taker market would have a lower nominal spread and the taker/maker market would have a higher nominal spread, but the cum fee bid-ask spread is the same for both markets (Angel, Harris, and Spatt, 2011; and Colliard and Foucault, 2012). However, SEC rule 611 states that orders should be routed to the market with the best displayed (nominal) spread. In that case, all orders should be routed to the maker/taker market and the taker/marker market should be empty. However, rule 612 prohibits sub-penny pricing, so the maker/taker and taker/maker markets can display the same nominal bid-ask spread. In addition, rule 611 imposes price priority only across markets, but time priority is imposed only on the individual market. Under tick-size constraints, the queue in the maker/taker market can be very long and order execution becomes the privilege of liquidity providers who trade at higher speeds. The taker/maker market provides a means of jumping ahead in the queue by paying a fee.

This paper is organized as follows. Section 2 describes the data used in the study. Section 3 examines the relationship between tick size constraints and HFT. Section 4 studies tick size constraints and taker/maker fees. Section 5 examines the impact of tick-size constraints on

⁸ On high-frequency trading, see Biais, Foucault, and Moinas (2011), Jovanovic and Menkveld (2010), Pagnotta and Philippon (2012), Chaboud, Chiquoine, Hjalmarsson, and Vega (2009), Hendershott and Riordan (2009 and 2011), Hasbrouck and Saar (2013), and Hendershott, Jones, and Menkveld (2013), among others. For maker/taker markets, see Foucault, Kadan, and Kandel (2013), Colliard and Foucault (2012), Brolley and Malinova (2012), Park and Malinova (2013), and Halmrast, Malinova, and Park (2013), among others.

⁹ For example, if the equilibrium spread without tick size is 0.3 cents on the maker/taker market and 0.8 cents on the taker/maker market, both markets will quote a one-cent spread due to price constraints.

liquidity as well as HFT and taker/maker fees using stock splits as an exogenous shock on relative tick size. Section 6 concludes the paper and discusses the policy implications.

2. Data

This paper uses four main datasets: a NASDAQ HFT dataset that identifies whether a liquidity maker/taker is an HFTer, daily TAQ data with a millisecond time stamp, NASDAQ TotalView-ITCH with a nanosecond time stamp, and CRSP.

The NASDAQ HFT dataset provides information on limit-order books and trades for 120 stocks selected by Hendershott and Riordan. The sample includes 40 large stocks from the 1000 largest Russell 3000 stocks, 40 medium stocks from stocks ranked from 1001–2000, and 40 small stocks from Russell 2001–3000. Among these stocks, 60 are listed on the NASDAQ and 60 are listed on the NYSE. Because the sample was selected in early 2010, three stocks actually disappear in our sample period so we have 117 stocks in our sample. The limit-order book data offer one-minute snapshots of the book with an indicator that breaks out liquidity providers into HFTers and non-HFTers. This enables us to examine the best quotes and depth provided by HFTers and non-HFTers. The trade file provides information on whether the traders involved in each trade are HFTers or non-HFTers. In particular, trades in the dataset are categorized into four types, using the following abbreviations: "HH": HFTers who take liquidity from other HFTers; "HN": hFTers who take liquidity from hon-HFTers who take liquidity from ther non-HFTers.

The consolidated trades file of daily TAQ data provides information on execution across separate exchanges for trades greater than or equal to 100 shares (O'Hara, Yao, and Ye, 2013). We use such data to calculate EDGA's market share relative to that of EDGX. In our sample

period, EDGX, like most exchanges, has a maker/taker fee structure whereby liquidity demanders pay a fee of 0.30 cents per share while liquidity providers get a rebate of 0.26 cents per share; EDGA has a taker/maker (or inverted) fee structure whereby liquidity suppliers pay a fee of 0.025 cents per share while liquidity demanders get a rebate of 0.015 cents per share.

The results we use to examine the cross-sectional variation in HFT and taker/maker fees are based on trades involving the 117 stocks in October 2010. NASDAQ high-frequency data provide the market shares of high-frequency liquidity provision for the 117 stocks for 2008–2009, February 22–26, 2010 and October 2010. EDGA and EDGX volumes are included in the TAQ data from July 2010. Therefore, we have measures of both high-frequency liquidity provision and market shares of the taker/maker market relative to that of the maker/taker market for October 2010. The summary statistics on these stocks are presented in Panel A of Table 1.

Insert Table 1 about Here

We also study the impact of relative tick size on liquidity, HFT, and the taker/maker market using stock splits as exogenous shocks to nominal stock prices, but these 117 shocks do not provide a large enough sample of splits. We examine all NYSE and NASDAQ firms that declared a two-for-one or greater stock split between January 2010 and November 2011 in the CRSP universe. Each of our pre- and post-event windows is comprised of the 30 trading days immediately before the stock-splitting date and the 30 trading days immediately after the stock-splitting date, including the splitting date. We exclude stocks that split more than once during the sample period. Among these stocks, 86 firms list trading data in the ITCH dataset. The summary statistics on these stocks are presented in Panel B of Table 1. Because the data on EDGA and EDGX is available only for trades occurring after July 1, 2011, we have 66 firms with data on EDGA and EDGX volumes.

We do not have high-frequency liquidity-provision data for these split stocks. Fortunately, Foucault, Kadan, and Kandel (2013) provide a proxy for high-frequency liquidity-making. In their model, an increase in tick size increases liquidity makers' profits. As a consequence, the monitoring intensity of liquidity-makers increases. This reduces time in the liquidity-making cycle, which in turn reduces the time gap between an execution of a limit order and the submission of another limit order. This time gap is measured using NASDAQ TotalView-ITCH data, which is a series of messages that describe orders added to, removed from, or executed on the NASDAQ. We also use ITCH data to construct a limit-order book with nanosecond resolution, which is the foundation for calculating liquidity. In particular, using NASDAQ TotalView-ITCH data allows us to construct the depth measure away from the best bid and ask. This is important for comparing the depth level before and after a split. For example, the depth within 20 cents of the best bid and offer for a stock with \$20 is equivalent to the depth within 10 cents of \$10 after a two-for-one split.

3. Tick Size Constraints and High-Frequency Liquidity Provision

This section argues that high-frequency liquidity provision is a consequence of tick size constraints. Tick size plays a central role in separating price competition from speed competition. For example, suppose there are three liquidity providers, one of whom (trader A) is willing to provide liquidity at 0.1 cents, another of whom (trader B) is willing to provide liquidity at 0.5 cents, and a third of whom (trader C) is willing to provide liquidity at 1 cent. When tick size is smaller or equal to 0.5 cents, trader A has price priority over traders B and C. When tick size is more than 0.5 cents but smaller than 1 cent, both traders A and B are willing to offer liquidity at the same constrained price, and the priority between A is B is determined by time. When tick

size is greater than or equal to 1 cent, all three traders quote the same price and it is their speed in providing liquidity that determines whose order is executed first. Therefore, a large tick size dilutes the impact of the trader who is able to quote the best price.. Meanwhile, a large tick size increases the importance of speed competition.

We first formalize our argument by examining the relative market share of HFTers and non-HFTers in terms of volume. We find that HFTers take a higher market share for low-priced stocks with market caps. The analysis of quotes and depth provide more details about the economic mechanisms that drive the results on volume.

3.1. Tick Size Constraints Measure

An intuitive measure of tick size constraints is price. Because stocks with prices above one dollar have a tick size of one cent, lower-priced stocks have a large relative tick size. Also, tick-size constraints are more likely to be binding for large stocks, for which the equilibrium bidask spread can be below one tick.

Our empirical study design is justified by Benartzi, Michaely, Thaler, and Weld (2009), who argue that nominal share prices are exogenous with respect to firm fundamentals other than the market cap.¹⁰ Baker, Greenwood, and Wurgler (2009) posit a catering theory of nominal stock prices, according to which firms split when investors place higher valuations on low-priced firms, and vice versa. However, the catering theory focuses more sharply on time-series

_

¹⁰ Their paper states that the nominal share price is a puzzle because it cannot be explained by the marketability hypothesis, the pay-to-play hypothesis, or signaling. The marketability hypothesis states that low-priced stocks are more attractive to individual investors. (Baker and Gallagher, 1980; Baker and Powell, 1993; Fernando, Krishnamurthy, and Spindt, 1999 and 2004; Lakonishok and Lev, 1987; and Byun and Rozeff, 2003). The pay-to-play hypothesis posits that firms can split their stocks to achieve optimal relative tick size. A higher relative size motivates more dealers to make markets and investors to provide liquidity by placing limit orders, despite its placing a high floor on the quoted bid-ask spread (Angel, 1997). The signaling hypothesis (Brennan and Copeland, 1988; Lakonishok and Lev, 1987; and Kalay and Kronlund, 2013) states that insiders use stock splits to signal information.

variations in stock prices while our analysis focuses on cross-sections. Campbell, Hilscher, and Szilagyi (2008) find that prices may predict distress risk when they are very low, but the same paper also acknowledges that such a prediction no longer applies when the price rises above \$15. In summary, prior literature indicates that cross-sectional variations in nominal stock prices are orthogonal to firm fundamentals other than the market cap. Therefore, we use price and market cap as our measure of tick size constraints.

3.2 Tick Size Constraints and Volume

NASDAQ high-frequency data indicate, for each trade, the maker and taker of liquidity. We are interested in the percent of volume with high-frequency liquidity provision. The original 120 stocks selected by Hendershott and Riordan include 40 large stocks from the 1000-largest Russell 3000 stocks, 40 medium stocks from stocks ranked from 1001–2000, and 40 small stocks from Russell 3000 stocks 2001–3000. A natural way to conduct the analysis is to sort the stocks 3 by 3 based on the market cap and the price level of the stock. We then sort the 117 stocks first into small, medium, and large groups based on the average market cap of September 2010, and each group is further subdivided into low, medium, and high sub-groups based on the average closing price of September 2010.

Suppose NH_{it} , HH_{it} , HN_{it} , and NN_{it} are the four types of share volume for stock i on each day t. For each portfolio J, the volume share with HFTers as liquidity providers relative to total volume is defined as:

$$HFT liquidity volumes hare_{J} = \frac{\sum_{i \in J} \sum_{t=1}^{T} (NH_{it} + HH_{it})}{\sum_{i \in J} \sum_{t=1}^{T} (NH_{it} + HH_{it} + HN_{it} + NN_{it})}$$
 (1)

Table 2 demonstrates that the volume with HFTers as liquidity providers decreases monotonically with stock prices and the market cap. For large-cap stocks, 49.29% of the volume is due to HFTers as liquidity providers for low-priced stocks; the number decreases to 38.48% for medium-priced stocks and further decreases to 35.53% for high-priced large-cap stocks. For low-priced stocks, 39.15% of the volume is from HFTers as liquidity providers for mid-cap stocks, while the number is only 23.40% for small-cap stocks. The volume results suggest that HFTer liquidity provision is indeed more active in stocks with high tick size constraints. Next, we analyze the economic mechanism that drives this cross-sectional variation by examining the quoting strategy of HFTers and non-HFTers.

Insert Table 2 about Here

3.3 Tick Size Constraints, Best Quotes, and Depth

This section provides the economic mechanism that drives the results on volume. Our main finding is that non-HFTers enjoy a price advantage. They are more likely to quote better prices than HFTers. As relative tick size decreases, non-HFTers are more likely to quote better prices than HFTers, thereby establishing price priority. When relative tick size is large, however, HFTers and non-HFTers are more likely to quote the same constrained price, which suggests that time determines the priority for providing liquidity. Because HFTers are less likely to quote better prices but still enjoy a speed advantage when price competition is constrained by tick size, a relatively large tick size favors HFTers and increases their market share.

NASDAQ high-frequency book data provides one-minute snapshots of the limit order book. At each ask and bid price, the data indicates the depth provided by both HFTers and non-HFTers. For each stock on each day, there are 391 best ask prices and 391 best bid prices. Our

analysis starts by dividing the best price (bid or ask are treated independently) into three types: 1) both HFTers and non-HFTers display the best price, 2) only HFTers display the best price and 3) only non-HFTers display the best price. Next, we aggregate the number of observations into each category for all stocks and dates in each portfolio.

Insert Table 3 about Here

Table 3 reveals that non-HFTers enjoy a price advantage over HFTers, and the advantage increases when relative tick size decreases. A comparison between columns 1 and 2 reveals that non-HFTers are more likely to display the best price than HFTers for eight of these nine categories. This result is very surprising because there are a number of theoretical and empirical results suggesting that HFTers are more likely to quote better prices than non-HFTers, either because they can minimize adverse selection cost (Hendershott, Jones and Menkveld, 2011) or because they can better manage their inventory cost (Brogaard, Hagstromer, Norden and Riordan, 2013). More importantly, non-HFTers are even more likely to quote better prices when relative tick size decreases. Column 4 demonstrates the ratio of the best price from HFTers relative to the best price from non-HFTers. For large- and medium-cap stocks, the ratio is a decreasing function of relative tick size. For example, for large-cap stocks with medium relative tick size, non-HFTers offer better prices 20.1% of the time and HFTers offer better prices 13.1% of the time, resulting in a ratio of 1.53 (20.1%/13.1%). For large stocks with high prices, the incidence of best prices being displayed by HFTers increases slightly to 16.6%, but that of best prices being displayed by non-HFTers increases dramatically to 38.8%. Therefore, the ratio of Non-HFTers as unique providers of best prices increases dramatically to 2.34 (38.8%/16.6%). Taken together, non-HFTers are more likely to provide better prices than HFTers, increasing the incidence of non-HFTers' quoting the best price as relative tick size decreases. Therefore, nonHFTers enjoy a price advantage over HFTers. When relative tick size is small, non-HFTers can quote better prices than HFTers and achieve price priority. This explains why the volume of liquidity provision from non-HFTers increases when relative tick size decreases.

When relative tick size or the market cap increases, however, HFTers become more and more likely to quote the same price as non-HFTers. This implies that price and market cap together is indeed a good proxy for tick size constraints, because time is needed to decide the priority between HFTers and non-HFTers. We still find that non-HFTers are more likely to offer better prices than HFTers, but large relative tick size constrains their ability to undercut the price. Non-HFTers offer better prices than HFTers 2.8% of the time, whereas HFTers offer better prices than non-HFTers 1.9% of the time. For 95.4% of the time, both HFTers and non-HFTers offer the same best price, and the speed advantage of HFTers implies that they will have priority over non-HFTers. Therefore, HFTers are less likely to quote better prices than non-HFTers do, but a large tick size allows them to quote the same price as non-HFTers. A large tick size, therefore, shifts the priority from non-HFTers to HFTers.

The results pertaining to best depth further confirm this intuition. We find that HFTers are more likely to be at the best depth for low-priced large-cap stocks. The depth data provide one-minute snapshots of the depth provided by HFTers and non-HFTers, {HFTdepth_{itm}, NonHFTdepth_{itm}}, where i is the stock, t is the date, and m is the time of day. The average depths provided by HFTers and non-HFTers for each stock on each day are:

$$HFTdepth_{it} = \frac{1}{M} \sum_{i=1}^{M} HFTdepth_{itm} \quad \text{and} \quad NonHFTdepth_{it} = \frac{1}{M} \sum_{i=1}^{M} NonHFTdepth_{itm} \quad (2)$$

The depth provided by HFTers relative to the total depth of portfolio J is then defined as:

¹¹ It can either because undercutting is too costly or because there is no room to undercut the price when the bid-ask spread is exactly a penny.

$$HFT depth share_{J} = \frac{\sum_{i \in J} \sum_{t=1}^{T} HFT depth_{it}}{\sum_{i \in J} \sum_{t=1}^{T} (HFT depth_{it} + NonHFT depth_{it})}$$
(3)

Table 4 reveals that the depth provided by HFTers decreases monotonically in price and market cap. For low-priced large-cap stocks, HFTers provide greater depth than non-HFTers (55.62% vs. 44.38%), whereas HFTers provide only 33.16% of such depth for high-priced large-cap stocks. The number decreases to 23.67% for high-priced mid-cap stocks, and further decreases to 21.83% for high-priced small-cap stocks.

Insert Table 4 about Here

In summary, we find that non-HFTers are more likely to quote better prices and achieve price priority when tick size is relatively small. When price competition is constrained to a greater extent by tick size, however, the priority moves to HFTers, who can quickly post orders at constrained prices. Therefore, we believe that tick size constraints facilitate HFTer liquidity provision.

4. Tick Size Constraint, HFT and Taker/Maker Market

We argue that the taker/maker market is another natural response to tick size constraints. It provides liquidity providers with a means of undercutting prices by paying the stock exchange (the maker fee), which allows the stock exchange to use part of the maker fee to subsidize liquidity takers. The market we examined in section 3 is a traditional maker/taker market, in which liquidity providers are paid. Three exchanges—Boston, BATS-Y, and EDGA—have inverted fee structures that charge liquidity providers and subsidize liquidity demanders. Two interesting questions immediately emerge. First, why are some liquidity providers paid whereas

others need to pay when providing liquidity for the same stock? Second, what forces determine the competition for order flow between these two markets?

We offer three hypotheses following the intuitions established in section 3. First, we conjecture that:

H1: HFTers are more likely to make the market in the maker/taker market, in which liquidity providing is paid. Non-HFTers are more likely to make the market in the taker/maker market, in which they need to pay to provide liquidity.

This (imperfect) separating equilibrium is generated through the comparative advantage enjoyed by HFTers. Holding all else equal, the unit profit obtained to make the market is higher in the maker/taker market conditional on execution, but such execution needs to be at the front of the queue. Non-HFTers do not have the speed advantage to move to the front of the queue. However, they can choose to pay a fee. Interestingly, we can observe how they jump ahead in the queue in terms of both price priority and time priority. Because each trading platform has its own time priority, traders at the back of a queue in the maker/taker market can be at the head of the queue in the taker/maker market by paying a fee. What is more, trading platforms charging liquidity providers usually subsidize liquidity demanders. If the nominal spread in the taker/maker market is the same as the maker/taker market, a liquidity demander with a smart router (Foucault and Menkveld, 2008) would go first to the taker/maker market because of the subsidy. Therefore, the taker/maker fee is another natural force in the price system that makes it possible to bypass tick size constraints.

Section 3 shows that a relatively large tick size constrains non-HFTers from undercutting HFTers. For stocks with larger tick size constraints, we expect that the taker/maker market plays a more important role in undercutting the price. When relative tick size decreases, it becomes

easier to undercut the price and weakens dependence on the taker/maker market, which brings us to our second hypothesis:

H2: The market share taken by Non-HFTers in the taker/maker market is high for low-priced stocks, decreasing as the stock price increases.

Finally, we conjecture that the volume in the taker/maker market relative to that in the maker/taker market is also an increasing function of relative tick size.

H3: EDGA volume relative to that of EDGX increases with relative tick size.

This conjecture is consistent with the tick size constraints hypothesis of Foucault, Kadan, and Kandel (2013), but is contrary to the agency hypothesis of Angel, Harris, and Spatt (2011) and Battalio, Corwin, and Jennings (2013). We will give a detailed explanation in section 4.2.

Next, we provide tests for these three hypotheses. We use the twin trading platforms EDGA and EDGX to inform our identification strategy. These two trading platforms have similar infrastructures with the major difference being in the breakdown of maker/taker fees. EDGA charges liquidity makers 0.025 cents per share whereas it provides a 0.015 cent rebate to liquidity takers. EDGX provides liquidity makers with 0.26 cents per share but charges liquidity takers 0.3 cents per share. Therefore, the competition for order flow between these two trading platforms can be explained only by differences in fee structure.

4.1. HFTers' Activity in the Taker/maker Market Relative to that in the Marker/taker Market

The TAQ data do not provide an identifier for HFTers. We use two commonly known measures for HFTer activity from TAQ data: the quote-to-trade ratio (Angel, Harris, and Spatt, 2013) and negative dollar volume divided by total number of messages (Hendershott, Jones, and

Menkeveld, 2011; Boehmer, Fong, and Wu, 2013). These are both relative measures. If there is an increase in activity on the part of non-HFTers relative to that of HFTers, both measures should decrease because HFTers are more likely to have a higher quote-to-trade ratio and a higher number of messages relative to dollar volume.

Because these two measures are proxies for HFTer activity, they are subject to some limitations. First, neither measure separates liquidity-providing HFTers from liquidity-demanding HFTers. However, Brogaard, Hagströmer, Norden, and Riordan (2013) show that liquidity-providing HFTers have a much higher order cancellation ratio than liquidity-taking HFTers. Because the quote-to-trade ratio and the total number of messages divided by trading volume are mainly driven by cancellations, we expect liquidity-making HFTers to be the main drivers of these two variables. Also, these two measures can also be affected by stock characteristics (O'Hara, Saar, and Zhong, 2013). Our empirical specification, however, controls for both stocks and time fixed effects and the comparison is made between stocks on the same day across two trading platforms. Denoting the HFTer measure for stock i on trading platform j on day t as HFT $_{ijt}$, we have:

$$\begin{aligned} \text{HFT}_{ijt} &= u_i + \gamma_t + \alpha + \beta_1 DummyEDGA_{ijt} + \beta_2 DummyEDGA_{ijt} * Prc_{-i} \\ &+ \beta_3 DummyEDGA_{ijt} * logmktcap_{-i} + \varepsilon_{ijt} \end{aligned} \tag{4}$$

Here u_i is the firm fixed effect, which controls for the fact that some firms undertake HFT activity. γ_t is the time fixed effect, which presupposes that some days have more cancellations or messages than other days. $DummyEDGA_{ijt}$ equals 1 if the quote-to-trade ratio is from EDGA and 0 otherwise. Prc_{-i} is the average price level of stock i in September 2010 minus the median average price of the 117-stock sample. The variable $logmketcap_{-i}$ is the log of the market cap of stock i in September 2010 minus the log of the median market cap of the

117 stocks in the sample. Both variables are normalized to facilitate the interpretation of β_1 .¹² Because of the controls for both firm and time fixed effects, the regression measures the difference between relative HFT activity in EDGA and relative HFT activity in EDGX. β_1 measures the overall difference between EDGA and EDGX in terms of relative HFTer activity, and β_2 and β_3 measures how the differences depend on the price level and market cap of the results.

Table 5 shows that EGDA experiences a relatively lower level of HFTer activity than EDGX. Column (1) shows that EDGA exhibits a much lower quote-to-trade ratio than EDGX. The constant term (EDGX) is 30.63 and the EDGA dummy is -9.35, which implies that for an firm with median price and median market cap, the quote-to-trade ratio of EDGX is 30.63 to 1 and the quote-to-trade ratio of EDGA is 21.28 (30.63-9.35) to 1. This result is consistence with hypothesis 1. Column (2) shows that the dollar volume to message ratio in EDGA is higher than EDGX by 0.69, which implies that EDGA have higher dollar volume to message than EDGX, which is also consistent with hypothesis 1.

Insert Table 5 About Here

 β_2 is positive and significant, which means that EDGA exhibits an even a lower level of relative HFT activity compared with EDGX when the stock price is low because β_1 is negative. An increase in the stock price, however, increases the relative level of HFT activity in EDGA compared with that in EDGX, which is consistent with hypothesis 2. For low priced stocks, Non-HFTers rely more on taker/maker market to undercut the price because of tick size constraints. Therefore, we observe that EDGA have a relatively larger non-HFT activity or relatively less HFT activity when stock price is low. As stock prices increases, non-HFTers rely less on

¹² Without this normalization, β_1 is interpreted as the coefficient for a stock with price 0 and market cap 0.

taker/maker market to undercut the price, which increase the relative level of HFT activity in EDGA compared with that in EDGX. Column (2) shows that the sign for market cap is also consistent with hypothesis 2. An increase in market cap increases tick size constraints. Therefore, non-HFTers rely more on the taker/maker market to undercut the price. Therefore, there is an increase in the volume-to-message ratio (or a decrease in the –volume-to-message ratio) in EDGA relative to EDGX because non-HFTers tend to trade at a higher volume for the same number of messages.

4.2 Volume of Taker/maker Market Relative to Marker/taker Market

Hypothesis 3 states that the volume of taker/maker relative to marker/taker increases in relative tick size. To show this, we first sort the 117 stocks 3 by 3 by average market cap and then by average price on September 2010. Next, we aggregate the EDGA and EDGX volumes for stocks in the portfolio for all days. EDGAratio is defined as the ratio of the aggregated EDGA volume divided by the aggregated EDGA volume plus the aggregated EDGX volume. Table 6 reveals two interesting patterns. First, the taker/maker market has a surprisingly large market share in stocks with large-cap stocks with high relative tick size. Large-cap low-priced stocks take 64.28% of the EDGA volume, implying that EDGX accounts for only 35.72% of the volume. The taker/maker market also has a higher market share relative to the maker/taker market for low-priced mid-cap stocks (55.94% vs. 44.06) and medium-priced large-cap stocks. Second, as stock prices increase, volume shifts from the taker/maker market to the maker/taker market. For example, EDGX beats EDGA in large-cap high-priced stocks. EDGA accounts for only 28.98% of the volume with the remaining 71.02% in EDGX. Therefore, the taker/maker fee market takes a relatively higher market share in low-priced stocks, whereas the maker/taker fee

market takes a relatively higher market share in high-priced stocks. This demonstrates that liquidity providers are more willing to pay a fee to make a market for low-priced stocks.

Our result is consistent with the seminal theoretical paper on maker/taker fees by Foucault, Kadan, and Kandel (2013). Their model posits an optimal bid-ask spread without tick size constraints, or a spread that maximizes trading volume. ¹³ Tick size constraints the adjustment to the optimal bid-ask spread, but exchanges can adjust maker/taker fees to achieve the optimal spread. When the mandated tick size is too high, charging liquidity makers and subsidizing liquidity takers can increase trading volume. Based on the intuition of the model, the fact that large stocks have higher volume in the taker/maker market relative to the maker/taker market implies that a one-penny tick size might be too high for these stocks. As stock prices increase, relative tick size decreases and we observe a migration of volume from the taker/maker market to the maker/taker market due to smaller tick size constraints.

Insert Table 6 about here

We are aware, however, that there exist a competing hypothesis for explaining the market share of the taker/maker market relative to that of the maker/taker market based on fee structure. The agency hypothesis proposed by Angel, Harris, and Spatt (2013) and Battalio, Corwin, and Jennings (2013) argues that brokerage firms have an incentive to route non-marketable limit orders from retail traders to the maker/taker market because retail traders usually do not claim the rebate. The agency hypothesis is, however, unlikely to explain the cross-sectional variation of market shares in the taker/maker market relative to the maker/taker market. Existing empirical evidence either argues that retail traders are more likely to trade low-priced stocks (Baker and Gallagher, 1980; Baker and Powell, 1993; Fernando, Krishnamurthy, and Spindt, 1999 and

¹³ Page 316, equation (22) of *Journal of Finance*, February, 2013.

2004) or retail traders are indifferent between high-priced and low-priced stocks (Lakonishok and Lev, 1987; Benartzi, Michaely, Thaler, and Weld, 2009). In the first case, the agency hypothesis predicts that the maker/taker market should have higher market share in low-priced stocks. The agency hypothesis yields no prediction pertaining to cross-sectional variation in market share when retail traders are indifferent between low-priced and high-priced stocks. We find that the maker/taker market is more active for high-priced stocks, which is not consistent with the agency hypothesis.

Because high-frequency liquidity provision and the taker/maker market are driven by the same common factors, they are highly correlated. We first calculate the average percentage of BBO depth provided by high frequency traders for each stock, and then sort the 117 stocks into five quintiles based on the percentage. Stocks in quintile 1 exhibit the lowest percentage of the BBO depth provided by high frequency traders, and stocks in quintile 5 exhibits the highest percentage of BBO depth provided by high frequency traders. For each stock in the quintile, we find the average percentage of the EDGA trading volume relative to the total volume of EDGA and EDGX, and we equally weight the stocks in each quintile. Figure 1 demonstrates that the portfolio with the highest HFT liquidity provision (quintile 5) has the highest market share in the taker/maker market (56%). The market share of the taker/maker market decreases monotonically with HFT liquidity provision. Quintile 1, or the stock with lowest HFT liquidity provision, only has 28% of volume in taker/maker market.

Insert Figure 1 about Here

5. The Impact of Tick Size Constraints on Liquidity

We engage with recent policy debates over whether an increase in tick size improves liquidity (SEC, 2012) by using a stock-splitting event as an exogenous shock to relative tick size. Following stock splits, outstanding shares are multiplied while nominal stock prices are reduced by the same factor. Therefore, tick size constraints become more binding after stock splits. In addition, we use the increase in relative tick size after splits as a robustness check for our results on cross-sectional variations in HFT and taker/maker market share.

The HFT dataset we used in previous sections cover only 117 stocks and only one of those stocks has experienced a stock split. In order to study the general trend, we examine all firms that declared a two-for-one, three-for-one, or four-for-one stock split between January 2010 and November 2011 in the CRSP universe. Each of our pre- and post-event windows is comprised of the 30 trading days immediately before the stock-splitting date and the 30 trading days immediately after the stock-splitting date, including the splitting date. We exclude stocks that split more than once during the sample period. Among these stocks, 86 firms list trading data in the ITCH dataset and 66 list data on EDGA and EDGX volumes. ¹⁴ To address potential issues regarding the time trend in our sample period, we also match splitting stocks one-to-one with stocks that do not split based on price, market cap, and listing exchange. Therefore, for each stock that splits, we match it with a stock listed on the same exchange with minimal matching error D_{ii}, where the matching error is defined as:

$$D_{ij} = \left| \frac{MCAP_{i}}{MCAP_{i}} - 1 \right| + \left| \frac{PRC_{i}}{PRC_{i}} - 1 \right|$$
(5)

 $^{^{14}}$ We have fewer data on EDGA and EDGX volumes because the data were not available from January 2010–June, 2010.

Section 5.1 examines the impact of splits on liquidity, and section 5.2 examines the impact of splits on HFT and taker/maker volume. Section 5.3 provides a brief discussion to explain the economic linkage between the results reported in section 5.1 and those reported in section 5.2.

5.1. Tick Size Constraints and Liquidity

We explore the relation between tick size constraints and liquidity using the diff-in-diff approach. We run the following regression:

Liquidity_{it} = $\alpha + \beta_1 treatment_i + \beta_2 after_t + \beta_3 treatment_i * after_t + e_{it}$ (6) where $treatment_i$ is equal to 1 for stocks that split and 0 for the matched sample, $after_t$ is equal to 1 after the splitting day for stock 1 and 0 before the split. The variable of interest is β_3 , which measures the impact of the tick size constraints on liquidity. Stock market liquidity is defined as the ability to trade a security quickly at a price close to its consensus value (Foucault, Pagano, and Röell, 2013). Spread is the transaction cost faced by traders, and is often measured by the quoted bid-ask spread or the trade-based effective spread. Depth reflects the market's ability to absorb large orders with minimal price impact, and is often measured by the quoted depth.

The results are presented in Table 7. Panel A presents the analysis of the impact of changes in tick size constraints on the quoted spread. The dependent variable in each column is defined as follows. *pQspread* in Column (3) is the time-weighted proportional quoted spread for each stock on each day, where proportional quoted spread is defined as the quoted spread divided by the midpoint of the best ask and bid. In addition to earning the quoted spread, a market maker

also obtains a rebate from each executed share from the exchange. Therefore, we compute two other measures of the quoted spread. $Qspread_{adj}$ in Column (2) and $pQspread_{adj}$ in Column (4) are the spreads adjusted by the liquidity supplier's rebate. Specifically,

$$Qspread_{adj_{it}} = Qspread_{it} + 2 * liquidity maker rebate$$
 (7)

$$pQspread_{adj_{it}} = (Qspread_{it} + 2 * liquidity maker rebate)/midpoint_{it}$$
 (8)

Insert Table 7 About Here

Panel B presents the analysis of the impact of changes in tick size constraints on Nasdaq limit-order book depth. The variable *Depth* in Column (1) measures the sum of the shares at the best bid and ask prices except for the splitting firms before the stock splits. The variable *Depth10* in Column (2) measures the sums of shares at and within 10 cents from the best bid and ask prices except for the splitting firms before the stock splits. Because splits cause an increase in shares and a decrease in share price, we make the following adjustments, following Lipson (1999), for the depth prior to the split. The basic idea is to measure the depth within a certain percentage of the stock price. Therefore, *Depth* for splitting firms before the splitting event is the sum of the shares at and within one, two, and three cents from the best quote for two-for-one splits, three-for-one splits, and four-for-one splits, respectively. **In Depth10** The presents splitting firms before splitting events as the sum of the orders at and within twenty, thirty, and forty cents from the best quote for two-for-one splits, three-for-one splits, and four-for-one splits, respectively. After the stock split, tick size constraints become more binding as nominal prices become a fraction of the pre-splitting level.

¹⁵ For each stock on each day, the liquidity maker's rebate is 0.295 cent per execution, but the results are qualitatively similar with other rebate levels.

¹⁶ We also measure *Depth* for the splitting firms before the splitting event as the best quote and the result is stronger,

¹⁶ We also measure *Depth* for the splitting firms before the splitting event as the best quote and the result is stronger, but for the sake of brevity we do not present it.

Panel C presents the analysis of the impact of changes in tick size constraints on the effective spread. The dependent variable in each column is defined as follows. The effective spread for a buy is defined as twice the difference between the trade price and the midpoint of the best bid and ask price. The effective spread for a sell is defined as twice the difference between the midpoints of the best bid and ask and the trade price. *Espread* in Column (1) is defined as the size-weighted effective spread of all trades for each stock and each day. *Espread* in Column (2) is defined as the spreads adjusted by the liquidity taker's fee.

$$Espread_{adj_{it}} = Espread_{it} + 2 * liquidity taker fee$$
 (9)

The proportional effective spread is defined as the effective spread divided by the midpoint. pEspread in Column (3) is defined as the size-weighted proportional effective spread of all trades for each stock and each day. $pEspread_{adj}$ in Column (4) are spreads adjusted by the liquidity taker's fee. Specifically,

$$pEspread_{adj_{it}} = (Espread_{it} + 2 * liquidity taker fee)/midpoint_{it}$$
 (10)

Panel A shows that both the quoted spread and the adjusted quoted spread decrease by 5.86 cents after the split. This result is not surprising because a split results in a decrease in the nominal share price, and the quoted spread should decrease. However, the proportional quoted spread increases by 5 basis points and the adjusted quoted spread increases by 6.1 basis points. The adjusted proportional quoted spread increases more because of the fee. For example, suppose a \$20 stock has a two-for-one split and a \$2,000 transaction has a maker rebate of 29.5 cents (0.295 cents per share*100 shares), but the maker rebate increases to 59 cents after the split (0.295 cents per share *200 shares). Therefore, the maker fee causes a further increase in the proportional quoted spread after adjusting for the fee. Panel B shows that limit-order-book depth

increases after the stock splits. This holds true both for depth at the best bid and ask prices and for the accumulative depth within 10 cents of the best bid and ask prices.

With an increase in both the quoted spread and depth, the key variable of interest becomes the effective spread, because it measures the transaction cost to traders. Panel C demonstrates a decrease in the effective spread after the split. However, this decrease in the effective spread is not as large as the decrease in the share price. The proportional effective spread then increases, particularly after adjusting for the fees. This shows that the proportional transaction cost increases after the split. Therefore, an increase in relative tick size does not improve liquidity.

5.2. Tick Size Constraints, HFT Liquidity Providing and Take/Maker Market Share

In this section, we examine the impact of changes in tick size constraints on HFT market marking and taker/maker market share. The sample we use in this section is the same as the one used in the previous section, namely, firms that split stocks and their matching stocks.

For those stocks, we do not have high-frequency liquidity provision data at hand, so we are not able to directly observe HFTer liquidity provision. Fortunately, Foucault, Kadan, and Kandel (2013) provide a proxy for the speed of liquidity-making. In their model, an increase in tick size increases liquidity makers' profits. As a consequence, the monitoring intensity of liquidity-makers increases. This reduces time in the liquidity-making cycle, which reduces the time gap between an execution of a limit order and the submission of another limit order. Following this model, we estimate the speed competition of liquidity provision as follows: suppose that an order is executed with an E message; we then look for the closest order

submission with the same size and order side. Therefore, some resubmission time associated with each trade exists. Next, we calculate the 1st, 5th, 10th, 25th, 50th, 75th, 90th, 95th, and 99th percentiles of resubmission time for each stock on each day. Table 8 shows the correlation of the logarithm of the resubmission time with measures of high-frequency liquidity-making for the 117 stocks. The first column shows the correlation between resubmission time and the percentage of volume with HFTers as liquidity providers, and the second column shows the correlation between resubmission time and the percentage of depth provided by HFTers. The 1st–75th resubmission times are negatively correlated with high-frequency liquidity-making, which means that a lower resubmission time, or a higher resubmission speed, is associated with a larger fraction of high-frequency market-making. The correlation becomes insignificant from the 90th percentile to the 99th percentile. Therefore, we chose the 1st–50th resubmission time percentiles as our proxy for speed competition in liquidity making.

Insert Table 8 About Here

Next, using the diff-in-diff approach, we explore the effects of changes in tick size constraints on speed competition in liquidity making and on the market share of the taker/maker market. For speed competition, we run the following regression:

Speed_{it} =
$$\alpha + \beta_1 treatment_i + \beta_2 after_t + \beta_3 treatment_i * after_t + e_{it}$$
 (11)
where $treatment_i$ is equal to 1 for stocks that split and 0 for the matched sample, $after_t$ is equal to 1 after the splitting day for stock 1 and 0 before the split. The variable of interest is β_3 , which measures the impact of the split on trading speed.

We run a similar regression on EDGA's market share relative to EDGX (Edgeratio) as the dependent variable:

 $Edgeratio_{it} = \alpha + \beta_1 treatment_i + \beta_2 after_t + \beta_3 treatment_i * after_t + e_{it}$ (12) where $Edgeratio_{it}$ is defined as the EDGA volume divided by the sum of the EDGA and EDGX volumes.

The results are presented in Table 9. The first five columns present the results on speed. The constant term is the trading speed of the control group before the treatment. For example, the 1^{st} percentile of the resubmission time is exp (-7.81) = .000406 seconds and the 50^{th} percentile of the resubmission time is exp (-0.951) = 0.386 seconds. The time scale of the resubmission is too short to have been submitted by humans. Nor would the change in time be driven by economic fundamentals. We argue that the change is more likely to be driven by speed competition among HFTers. The coefficient β_1 is not significant for the 1^{st} – 25^{th} percentiles and only marginally significant for the 50^{th} percentile, meaning that the order resubmission time in the treatment group is very similar to the resubmission time of the control group before the split. This result suggests that price, market cap, and listing exchange provide an appropriate control group for trading speed. The variable *after*_t is not significant, implying that there is no general trend regarding trading speed.

Insert Table 9 About Here

The coefficient of interest is β_3 , which measures the average treatment effect of the stock split on order-resubmission time. Because the dependent variable representing speed is expressed in logarithmic format, the coefficient can be interpreted approximately as the percentage change in the resubmission time of the treatment group relative to that of the matched stocks that do not split. For example, the 1st percentile of the resubmission time of the split stocks decreases by 19.5% after the split, while the 50^{th} percentile of the resubmission time

decreases by 29.3%. The results show that, after the stock split, there is a dramatic increase in order-resubmission speed. The sixth column demonstrates the results for EDGEratio. We find that Edge A's volume increases by 7.32% relative to that of Edge X.

5.3. Summary

In summary, the results based on a stock-splitting event indicate that an increase in tick size constraints does not improve liquidity. However, the constrained price spurs speed competition because of time priority, while also resulting in an active taker/maker market.

6. Conclusion

We argue that tick size regulation is among the drivers of two important features of current stock markets: HFT and taker/maker markets. Because rule 612 of regulation NMS prohibits price competition at increments less than one cent, speed allocates executions when price competition is constrained. Moreover, the one-cent tick size applies to all stocks priced greater than one dollar, which implies heterogeneous tick size constraints for stocks with different market caps and price levels. We find that non-HFTers have a comparative advantage in supplying liquidity for stocks with small relative tick sizes because of their ability to quote better prices. A decrease in stock prices or an increase in relative tick size, however, favors HFTers. Higher tick size constraints eliminate non-HFTers' ability to undercut HFTers. Therefore, both HFTers and non-HFTers quote the same prices and HFTers can achieve priority because of their speed advantage. Stocks with higher tick size constraints also take larger market shares in the taker/maker market, in which the price constraint can be bypassed by paying a fee.

We also find that a decrease in the nominal share price due to a stock split is associated with an increase in trading speed and a migration of volume from the maker/taker market to the taker/maker market. These results, again, suggest that tick size constraints and time priority are the two common factors that drive speed competition and the taker/maker market.

Currently, regulators in the United States and Europe are considering increasing tick sizes with the following two arguments. First, a larger tick size increases liquidity, facilitates stock research, and ultimately increases IPO. Second, a larger tick size controls HFT. We argue that none of this argument is justified. We find that an increase in relative tick size does not improve liquidity. Instead, it increases competition in speed by constraining price competition. The argument that price constraints lead to non-price competition is valid, but we doubt whether non-price competition will involve publishing stock research. A more direct way to create non-price competition is speed competition. Also, an increase in tick size constraints would lead to a migration of volume to the taker/maker market, in which tick size constraints can be bypassed.

Our paper can be extended in various ways. First, current theoretical work on speed competition focuses on the role of information. Our paper points out another channel for speed competition: tick size constraints. Models using discrete prices can be constructed to indicate the value of speed and the impact of tick size constraints on market quality. Second, we explain the market share of taker/maker fees based on tick size constraints, and theoretical models can be built to understand why there exist separate equilibria for traders on separate trading platforms and how exchanges set fee structures in both markets to maximize total profits. Empirically, the relationship between tick size constraints, speed competition, and maker/taker versus taker/maker fees can be further explored. For example, the SEC recently announced a pilot

program for increasing tick sizes for a number of small stocks, and it would be interesting to see the impact of this shock on speed competition and the taker/maker fee market.

REFERENCES

- Angel, J., 1997, Tick Size, Share Prices, and Stock Splits, Journal of Finance 52, 655–681.
- Angel, James, Harris, Lawrence and Spatt, Chester S., 2010, Equity Trading in the 21st Century. Marshall School of Business Working Paper No. FBE 09-10.
- Angel, James, Harris, Lawrence and Spatt, Chester S., 2013, Equity Trading in the 21st Century: An Update. Marshall School of Business Working Paper.
- Avramovic, Ana, 2012, Manufacturing Volume: The Stock Split Solution, Credit Suisse, available from Edge, accessed Dec. 18, 2013.
- Baker, H., and P.L. Gallagher, 1980, Management's View of Stock Splits, *Financial Management* 9, 73–77.
- Baker, H., and G. Powell, 1993, Further Evidence on Managerial Motives for Stock Splits, *Quarterly Journal of Business and Economics* 32, 21–31.
- Baker, M., Robin Greenwood and Jeffrey Wurgler, 2009, Catering through Nominal Share Prices, *Journal of Finance* 64, 2559-2590.
- Bartlett, R. P., and J. McCrary, 2013, Shall we haggle in pennies at the speed of light or in nickels in the dark? How minimum price variation regulates high frequency trading and dark liquidity, Working paper, Berkeley School of Law.
- Battalio, Robert H., Corwin, Shane A. and Jennings, Robert H., 2013, Can Brokers Have It All? On the Relation between Make Take Fees & Limit Order Execution Quality, Working Paper, University of Notre Dame and Indiana University Bloomington.
- Benarzi, Shlomo, Roni Michaely, Richard Thaler and William C. Weld, 2009, The Nominal Share Price Puzzle, *Journal of Economic Perspectives* 23, 121-142.
- Biais, Bruno, Thierry Foucault and Sophie Moinas, 2013, Equilibrium High-Frequency Trading, Working paper, Toulouse School of Economics, HEC Paris and Toulouse School of Economics.
- Boehmer, Ekkehart, Kingsley Y.L. Fong and Juan Wu, 2012, International Evidence on Algorithmic Trading, Working Paper, EDHEC Business School, University of Georgia and University of New South Wales.
- Brennan, J. Michael and Thomas E. Copeland, 1988, Stock splits, stock prices, and transaction costs, *Journal of Financial Economics* 22, 83-101.
- Brolley, Michael and Katya Malinova, 2012, Informed trading and maker-taker fees in a low-latency limit order market, Working paper, University of Toronto.

- Brogaard, Jonathan, Björn Hagströmer, Lars L. Norden and Ryan Riordan, 2013, Trading Fast and Slow: Colocation and Market Quality, working paper.
- Brogaard, Jonathan, Terrence Hendershott and Ryan Riordan, 2013, High frequency trading and price discovery, *Review of Financial Studies* forthcoming.
- Buti, S., Francesco Consonni, B. Rindi and I.M. Werner, 2013, Sub-Penny and Queue-Jumping, working paper, University of Toronto, Bocconi University and Ohio State University.
- Buti, S., B. Rindi, Y. Wen, and I. M. Werner, 2013, Tick size regulation and sub-penny trading, Working paper, University of Toronto, Bocconi University and Ohio State University.
- Budish, Eric, Cramton, Peter and Shim, John, 2013, The High-Frequency Trading Arms Race: Frequent Batch Auctions as a Market Design Response, working paper, University of Chicago and University of Maryland.
- Byun, J. and M.S. Rozeff, 2003, Long-run Performance after Stock Splits: 1927 to 1996, *Journal of Finance* 58, 1540–6261.
- Campbell, John Y., Jens Hilscher and Jan Szilagyi, 2008, In search of distress risk, *Journal of Finance* 63, 2899-2939.
- Chaboud, Alain, Benjamin Chiquoine, Erik Hjalmarsson, and Clara Vega, 2009, Rise of the machines: Algorithmic trading in the foreign exchange market, Working paper, Board of Governors of the Federal Reserve System.
- Colliard, Jean-Edouard and Foucault Thierry, 2012, Trading fees and efficiency in limit order markets" *Review of Financial Studies*, 25, 3389–3421.
- Fernando, C., S. Krishnamurthy and P. Spindt, 1999, Is Share Price Related to Marketability? Evidence from Open-end Mutual Fund Share Splits, *Financial Management* 28(3), 15–31.
- Fernando, C., S. Krishnamurthy and P. Spindt, 2004, Are Share Price Levels Informative? Evidence from the Ownership, Pricing, Turnover, and Performance of IPO Firms, *Journal of Financial Markets* 7, 377–403.
- Foucault, Thierry, Marco Pagano and Alisa Röell, 2013, *Market Liquidity*, Oxford University Press.
- Foucault, Thierry, Ohad Kadan and Eugene Kandel, 2013, Liquidity cycles, and make/take fees in electronic market, *Journal of Finance* 68, 299–341.
- Foucault, Thierry and Albert J. Menkveld, 2008, Competition for order flow and smart order routing systems, *Journal of Finance* 63, 119-158.

- Hasbrouck, Joel, and Gideon Saar, 2009, Technology and liquidity provision: the blurring of traditional definitions, *Journal of Financial Markets* 12, 143–172.
- Hasbrouck, Joel and Gideon Saar, 2013, Low-latency trading, *Journal of Financial Markets* forthcoming.
- Hendershott, Terrence, and Ryan Riordan, 2011, Algorithmic trading and information, Working paper, University of Washington and University of California, Berkeley.
- Hendershott, Terrence, Charles M. Jones, and Albert J. Menkveld, 2011, Does algorithmic trading improve liquidity? *Journal of Finance* 66, 1–33.
- Hirschey, Nicholas H., 2012, Do High-Frequency Traders Anticipate Buying and Selling Pressure? Working paper, London Business School.
- Halmrast, Nathan, Katya Malinova and Andreas Park, 2012, Does maker-taker pricing incentivize liquidity provision: evidence from the TSX, working paper, University of Toronto.
- Jones, Charles, 2013, What Do We Know about High-Frequency Trading? Working Paper, Columbia University.
- Jovanovic, Boyan, and Albert J. Menkveld, 2011, Middlemen in limit order markets, Working paper, VU University Amsterdam.
- Kalay, Alon and Mathias Kronlund, 2013, The market reaction to stock split announcement: earnings information after all, working paper, Columbia University and University of Illinois at Urbana-Champaign.
- Kavajecz, Kenneth A., 1999, A specialist's quoted depth and the limit order book, *Journal of Finance* 54, 747–771.
- Kirilenko, Andrei, Albert S. Kyle, Mehrdad Samadi and Tugkan Tuzun, 2011, The flash crash: The impact of high-frequency trading on an electronic market, Working paper, MIT Sloan School of Management, University of Maryland, University of North Carolina at Chapel Hill, and Federal Reserve Board.
- Lakonishok, J., and B. Lev, 1987, Stock Splits and Stock Dividends: Why, Who and When, *Journal of Finance* 42, 913–932.
- Marc Lipson, 1999, Stock splits, liquidity and limit orders. Working Paper, Darden School of Business

- Lo, Andrew W. and A. Craig MacKinlay, 1988, Stock market prices do not follow random walks: Evidence from a simple specification test, *Review of Financial Studies* 1, 41–66.
- Madhavan, Ananth, 2011, Exchange-Traded Funds, Market Structure and the Flash Crash Working paper, BlackRock, Inc.
- O'Hara, Maureen, Chen Yao, and Mao Ye, 2011, What's not there: The odd-lot bias in market data, Working paper, Cornell University, University of Illinois at Urbana-Champaign, and University of Illinois at Urbana-Champaign.
- O'Hara, Maureen, Gideon Saar and Zhuo Zhong, 2013, Relative Tick Size and the Trading Environment, working paper, Cornell University.
- Malinova, Katya and Andreas Park, 2013, Subsidizing liquidity: the Impact of make/take fees on market quality, *Journal of Finance* forthcoming.
- Patterson, Scott, 2012, Crown Business, New York.
- Pagnotta, Emiliano and Thomas Philippon, 2012, Competing on speed, Working paper, New York University.
- Rockoff, Hugh, 2008, Price Controls, *The Concise Encyclopedia of Economics*, Indianapolis: The Liberty Fund, 409-412.
- SEC, 2012, Report to Congress on Decimalization

Table 1 Sample Summary Statistics

This table presents summary statistics of the sample data used in the paper. Panel A presents summary statistics of the 117 stock sample as of October 2010 using CRSP market data. Panel B presents descriptive statistics of both the split sample and matched firms 30 days prior to the split execution day using both CRSP and NASDAQ (Total View-ITCH) data.

Panel A: CRSP Summary Statistics of 117 Stocks as of October 2010						
	Market Cap	Avg Closing Price	Avg Daily Volum	ne Avg Daily Return		
	(\$Million)	(\$)	(1000s)	(%)		
Mean	19352	41.5	5095	0.245		
Medium	2032	27.22	573	0.23		
Std	42246	64.41	11176	0.365		
Min	282	5.72	24	-0.665		
Max	275000	575.94	67028	1.331		
Panel B: Descriptive Statistics of Split Sample & Matched Firms						
		Mean		Median		
	N	Sample	Match 9	Sample Match		

		Me	ean	Med	lian
	N	Sample	Match	Sample	Match
Split Factor	86	2.36	-	2	-
Pre-split Price (\$)	86	84.14	65.19	71.76	54.44
CRSP					
Market Cap (\$Million)	86	7065	6130	3440	2146
Avg Closing Price (\$)	86	82.57	63.56	72.4	53.48
Avg Daily Volume (1000s)	86	917	1007	467	446
Avg Daily Return (%)	86	0.153	0.142	0.058	0.066
NASDAQ (Total View-ITCH)					
Avg Quoted Spread (\$)	86	0.123	0.097	0.096	0.08
Avg Effective Spread (\$)	86	0.048	0.04	0.029	0.023
Avg Best Depth	86	430	520	351	372
Avg Depth within 10 Cents from NBBO	86	3009	5708	1712	2591

Table 2: Percentage of Volume with High Frequency Traders as the Liquidity Providers by Market Cap and Relative Tick Size

This table presents the volume-weighted percentage of trading volume with high frequency traders as the liquidity providers. The sample includes 117 stocks in the NASDAQ high-frequency data in Oct 2010. The stocks are sorted first by average market cap and then by average price in September 2010 into 3 by 3 portfolios. To calculate the volume-weighted average, we first aggregate the volume with high-frequency traders as the liquidity providers and all volume in the portfolio for all days. Volume-weighted average is defined as the ratio of the volume with high-frequency traders as the liquidity providers divided by the aggregated overall volume.

	Large Relative Tick Size Medium Relative Tick (Low Price) (Medium Price)		Small Relative Tick Size (High Price)
Large Cap	49.29%	38.48%	35.53%
Middle Cap	39.15%	23.56%	22.34%
Small Cap	23.40%	19.93%	18.74%

Table 3: Who Provides the Best Quotes?

This table displays the percentage of time high frequency traders and non-high frequency traders provide the best bid and ask quotes to the NASDAQ limit order book. The sample includes 117 stocks in the NASDAQ high-frequency data in Oct 2010. The data provide a one-minute snap shot to the limit order book indicating the order size and the type of traders that submit the order at each price level. Stocks are sorted first into 3 by 3 portfolios by average market cap and then by average price in September 2010. We aggregate the number of times that high frequency traders, and non-high frequency traders are the sole providers of the best quotes, and the number of times that both type traders provide the best quotes for each stock category. We then divide the number of the best quotes associate with each category by the aggregate number of snap shots in that category. Column (1) present the percentage of time that high frequency traders is the sole provider for the best quotes, and Column (2) presents the percentage of time that non-high frequency trader is the sole providers for the best quotes. Column (3) presents the percentage of time that both high frequency traders and non-high frequency traders provide the best quotes. Column (4) is the ratio of column (2) to column (1).

		(1)	(2)	(3)	(4)
	Relative Tick Size	HFT	Non-HFT	HFT and Non-HFT	Ratio
	Large (Low Price)	1.9%	2.8%	95.4%	1.47
Large Cap	Medium (Medium Price)	13.1%	20.1%	66.9%	1.53
	Small (High Price)	16.6%	38.8%	44.7%	2.34
	Large (Low Price)	18.0%	15.2%	66.8%	0.84
Medium Cap	Medium (Medium Price)	20.0%	56.6%	23.4%	2.83
	Small (High Price)	20.7%	63.7%	15.7%	3.08
	Large (Low Price)	11.3%	54.7%	34.1%	4.84
Small Cap	Medium (Medium Price)	20.2%	55.8%	24.0%	2.76
	Small (High Price)	18.6%	70.7%	10.7%	3.80
Total		15.6%	42.1%	42.4%	2.70

Table 4: Market Share of Depth at BBO Provided by High Frequency Traders by Market Cap and Relative Tick Size

This table presents volume-weighted market shares of NASDAQ best bid and offer (BBO) depth provided by high-frequency traders. The sample includes 117 stocks in the NASDAQ high-frequency data in Oct 2010. The stocks are sorted first by average market cap and then by average price in September 2010 into 3 by 3 portfolios. To calculate the volume-weighted average, we first aggregate the depth provided by high-frequency traders and the depth provided by all traders for stocks in the portfolio for all days. Volume-weighted average is defined as the ratio of the aggregated depth provided by high frequency traders divided by the aggregated depth provided by all traders.

Large Relative Tick Size (Low Price)		Medium Relative Tick Size (Medium Price)	Small Relative Tick Size (High Price)
Large Cap	55.62%	46.17%	33.16%
Middle Cap	39.60%	30.26%	23.67%
Small Cap	24.92%	23.19%	21.83%

Table 5: Does Tick Size Constraints Affect the Relative High Frequency Trading Activities Between EDGA and EDGX?

This table presents the impact of tick size constrains on the relative high frequency trading activities between Direct Edge's two trading platforms, EDGA and EDGX. EDGA charges liquidity providers and pays rebates to liquidity takers, whereas EDGX charges liquidity takers and pays rebates to liquidity makers. We use the following regression spefication:

$$\begin{aligned} \text{HFT}_{ijt} &= u_i + \gamma_t + \alpha + \beta_1 DummyEDGA_{ijt} + \beta_2 DummyEDGA_{ijt} * Prc_{-i} + \beta_3 DummyEDGA_{ijt} \\ &* logmktcap_{-i} + \varepsilon_{ijt} \end{aligned}$$

We use two measures, <code>quote_to_trade_ratio</code> in Column (1), and <code>volume_to_message_ratio</code> in Column (2), as proxies for the high frequency trading activities. We measure <code>quote_to_trade_ratio</code> as the ratio between the number of quotes and the number of trades in each platform for each stock on each day. We measure <code>volume_to_message_ratio</code> as the negative of dollar trading volume (in \$100) divided by the number of messages, where message is calculated as the sum of the number of quote and the number of trade, in each platform for each stock on each day. <code>dummyEDGA</code> is a dummy variable that equals one if the HFT measure is from EDGA, and equals to zero if the HFT measure is for EDGX. <code>Prc_i</code> is the average price level of stock <code>i</code> in September 2010 minus the median average price of the 117-stock sample. The variable <code>logmktcap_i</code> is the log of the market cap of stock <code>i</code> in September 2010 minus the log of the median market cap of the 117 stocks in the sample. This table uses 117 stocks in Nsadaq High Frequency Trading Data. The dependent variable is constructed using TAQ data in Oct 2010. Standard errors are shown in parenthesis in the table; *, ** and *** represent statistical significance at 10%, 5% and 1% level respectively.

	(4)	(0)
	(1)	(2)
VARIABLES	Quote_to_trade_ratio	Volume_to_message_ratio
DummyEDGA	-9.350***	-0.690***
	(1.347)	(0.075)
DummyEDGA*prc	.0537***	.1353***
	(.0115)	(.014)
DummyEDGA*logmktcap	.8728	2035**
	(.5985)	(.092)
Constant	30.63***	-4.95***
	(2.809)	(.4082)
R-squared	0.227	0.784
Observations	4888	4888
Time effects	Yes	Yes
Firm effects	Yes	Yes

Table 6: The Market Share of Taker/maker Market by Market Cap and Relative Tick Size

This table presents the ratio of Direct Edge A (EDGA) volume to the total volume of Direct Edge (the volume of EDGA plus Direct Edge X (EDGX)). EDGA charges liquidity providers and pays rebates to liquidity takers, whereas EDGX charges liquidity takers and pays rebates to liquidity makers. The sample includes 117 stocks in the NASDAQ high-frequency data in Oct 2010. The 117 stocks are sorted first by market cap and then by price into 3 by 3 portfolios based on their average market cap and prices in September 2010. We aggregate the EDGA and EDGX volumes for stocks in each portfolio across all days. Volume weighted average is defined as the ratio of the aggregated EDGA volume divided by the aggregated EDGA volume plus the aggregated EDGX volume. Panel B is on equally weighted average. We first compute, for each stock i on each day t, the market share of EDGA relative to that of EDGA and EDGX. Then we average these observations equally across stocks and dates.

	Large Relative Tick Size (Low Price)	•	
Large Cap	64.28%	54.57%	28.98%
Middle Cap	55.94%	40.95%	38.33%
Small Cap	43.56%	29.79%	24.53%

Table 7: Impact of Tick Size Constraints on Market Liquidity

This table studies the impact of changes in tick size change on market liquidity. We use stock splitting event as an exogenous shock as the change of tick size constraints. We examine all firms that declared a two-for-one, three-for-one, or four-for-one stock split between January 2010 and November 2011 in the CRSP universe. Each of our pre- and post-event windows is comprised of the 30 trading days immediately before the stock-splitting date, and 30 trading days immediately after the stock-splitting date, including the splitting date. We excluded stocks that split more than once during the sample period. We employ the following diff-in-diff regression to study the relationship.

$$Liquidity_{it} = \alpha + \beta_1 treatment_i + \beta_2 after_t + \beta_3 treatment_i * after_t + e_{it}$$

Treatment_i is equal to 1 for stocks that split and 0 for the matched sample, after_t equal to 1 after the splitting day for stock 1 and 0 before the stock split. We measure the liquidity variable for each stock i on each day t. In Panel A column (1), we measure liquidity variable Depth as the sum of the bid and ask offers at Nasdaq BBO. To make fair comparison, Depth before the splitting event has been adjusted by the splitting ratio for the splitting stocks. In Column (2), variable Depth10 is measured as the sum depth at and within 10 cent of Nasdaq BBO. To make fair comparison, Depth10 before the splitting event has been adjusted by the splitting ratio for the splitting stocks. In Panel B, Qspread in Column (3) stands for quoted spread and $Qspread_{adj}$ in Column (4) stands for the quoted spread adjusted for the liquidity maker's rebate. pQspread in Column (5) is proportional quoted spread and $pQspread_{adj}$ in Column (6) is the proportional quoted spread and $dspread_{adj}$ in Column (8) stands for the effective spread in Column (7) stands for effective spread and $dspread_{adj}$ in Column (8) is effective spread and $dspread_{adj}$ in Column (9) is effective spread and $dspread_{adj}$ in Column (10) is the proportional effective spread and adjusted for liquidity taker's fee.

Panel A: The Impact of Change in Tick Size Constrains on Quoted Spread

	(1)	(2)	(3)	(4)
VARIABLES	Qspread	$Qspread_{adj}$	pQspread	$pQspread_{adj}$
after	0.00293	0.00293	9.30e-05	9.27e-05
	(0.00250)	(0.00250)	(0.000120)	(0.000122)
treatment	0.0296***	0.0296***	-0.000249**	-0.000298**
	(0.00250)	(0.00250)	(0.000120)	(0.000122)
treatment*after	-0.0586***	-0.0586***	0.000500***	0.000610***
	(0.00354)	(0.00354)	(0.000170)	(0.000173)
Constant	0.0935***	0.0994***	0.00223***	0.00238***
	(0.00177)	(0.00177)	(8.48e-05)	(8.65e-05)
Observations	9,182	9,182	9,182	9,182
R-squared	0.051	0.051	0.003	0.004

Panel B: The Impact of Change in Tick Size Constrains on Limit Order Book Depth

Constraint on Emili Order Book Beptil						
	(1)	(2)				
VARIABLES	Depth	Depth10				
after	50.33**	551.3**				
	(19.63)	(277.3)				
treatment	134.4***	-700.0**				
	(19.62)	(277.2)				
treatment*after	51.86*	3,497***				
	(27.76)	(392.2)				
Constant	505.4***	5,584***				
	(13.87)	(196.0)				
Observations	9,182	9,182				
R-squared	0.018	0.026				

Panel C: The Impact of Change in Tick Size Constrains on Effective Spread

	(1)	(2)	(3)	(4)
VARIABLES	Espread	$Espread_{adj}$	pEspread	$pEspread_{adj} \\$
after	0.000733	0.000733	2.33e-05	2.32e-05
	(0.00289)	(0.00289)	(9.86e-05)	(0.000101)
treatment	0.0194***	0.0194***	-4.07e-05	-9.09e-05
	(0.00289)	(0.00289)	(9.85e-05)	(0.000101)
treatment*after	-0.0477***	-0.0477***	0.000223	0.000336**
	(0.00409)	(0.00409)	(0.000139)	(0.000143)
Constant	0.0725***	0.0785***	0.00158***	0.00173***
	(0.00204)	(0.00204)	(6.97e-05)	(7.13e-05)
Observations	9,164	9,164	9,164	9,164
R-squared	0.029	0.029	0.001	0.002

Table 8: Correlation between High Frequency Market Making and Resubmission Time

This table presents the correlation between limit order resubmission time and high-frequency liquidity provision. We first compute the logarithm of the quickest i percentile of a limit order submission subsequent to an execution for stock i on date t, and then take the average across days. For each stock, we obtain logpi, with i = 1, 5, 10, 25, 50, 75, 95, 99 by taking the average across days. For each stock ,we compute HFT_Volume , the average percentage of trading volume with high frequency traders as the liquidity provider, and HFT_Depth , the average percentage of depth at Nasdaq BBO provided by high frequency traders. The sample includes 117 stocks in the NASDAQ high-frequency data in Oct 2010.

	HFT_Volume	HFT_Depth
logp1	-0.654***	-0.672***
P-Value	(0.000)	(0.000)
logp5	-0.605***	-0.630***
P-Value	(0.000)	(0.000)
logp10	-0.614***	-0.619***
P-Value	(0.000)	(0.000)
logp25	-0.649***	-0.645***
P-Value	(0.000)	(0.000)
logp50	-0.688***	-0.679***
P-Value	(0.000)	(0.000)
logp75	-0.517***	-0.678***
P-Value	(0.000)	(0.000)
logp90	0.028	-0.188**
P-Value	(0.765)	(0.044)
logp95	0.139	-0.045
P-Value	(0.138)	(0.634)
logp99	0.180*	0.086
P-Value	(0.053)	(0.358)

Table 9: Impact of Change in Tick Size Constraints on Speed Competition of Liquidity Making and Taker/maker Market Share

This table studies the effect of change in the tick size contains on HFT market making and on the market share of the taker/maker market. We use stock splitting event as an exogenous shock to the change of tick size constraints. We examine all firms that declared a two-for-one, three-for-one, or four-for-one stock split between January 2010 and November 2011 in the CRSP universe. Each of our pre- and post-event windows is comprised of the 30 trading days immediately before the stock-splitting date, and 30 trading days immediately after the stock-splitting date, including the splitting date. We excluded stocks that split more than once during the sample period. We employ the diff-in-diff regression to study the relationship. The specification for column 1-5 is

$$Speed_{it} = \alpha + \beta_1 treatment_i + \beta_2 after_t + \beta_3 treatment_i * after_t + e_{it}$$

where the proxies of the HFT market making speed is the 1^{st} , 5^{th} 10^{th} , 25^{th} and 50^{th} percentile of the resubmission speed of a limit order subsequent to an order execution. $Treatment_i$ is equal to 1 for stocks that split and 0 for the matched sample, after_t equal to 1 after the splitting day for stock i and 0 before the stock split.

The specification for column 6 is

$$Edgeratio_{it} = \alpha + \beta_1 treatment_i + \beta_2 after_t + \beta_3 treatment_i * after_t + e_{it}$$

where $Edgeratio_{it}$ is the volume of EDGA relative to the volume of EDGA plus EDGX. Standard errors are in parentheses, and ***, **, and * represent significance at the 1%, 5%, and 10% levels, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	logp1	logp5	logp10	logp25	logp50	edga_x
after	0.0237	0.0262	0.0669	0.108	0.00836	-0.0167
	(0.0710)	(0.0716)	(0.0755)	(0.0834)	(0.0799)	(0.0118)
pilot	-0.0232	-0.0166	-0.0892	-0.184**	-0.200**	-0.0477***
	(0.0699)	(0.0706)	(0.0744)	(0.0821)	(0.0787)	(0.0117)
pilot_after	-0.195**	-0.228**	-0.210**	-0.211*	-0.293***	0.0732***
_	(0.0991)	(0.100)	(0.105)	(0.116)	(0.112)	(0.0166)
Constant	-7.810***	-7.096***	-6.536***	-4.504***	-0.951***	0.310***
	(0.0500)	(0.0505)	(0.0532)	(0.0588)	(0.0564)	(0.00826)
Observations	9,169	9,169	9,169	9,169	9,169	6,556
R-squared	0.001	0.002	0.002	0.003	0.006	0.004

Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1

Figure 1: Market Share in Volume in the Taker/Maker Market with Respect to High Frequency Liquidity Provision

This figure shows relationship between the market share in trading volume in Direct Edge A and high-frequency liquidity provision in NASDAQ. The sample includes 117 stocks in the NASDAQ high-frequency data in Oct 2010. We first calculate the average percentage of BBO depth provided by high frequency traders for each stock, and then sort the stocks into five quintiles based on the percentage. Stocks in quintile 1 exhibit the lowest percentage of the BBO depth provided by high frequency traders, and stocks in quintile 5 exhibits the highest percentage of BBO depth provided by high frequency traders. For each stock in the quintile, we find the average percentage of the EDGA trading volume relative to the total volume of EDGA and EDGX, and we equally weigh the stocks in each quintile.

